Author’s response to reviews

Title: A prospective study of quality of life related to oral health after immediately loaded implant treatment in the atrophic edentulous maxilla, 3-year results

Authors:

Maria Erkapers (maria.erkapers@lul.se)
Susanna Segerström (susanna.segerstrom@lul.se)
Karl Ekstrand (karlekstrand@hotmail.com)
Russel Baer (rabaer@uadchicago.edu)
Joseph Toljanic (jtoljanic@midwestern.edu)
Andreas Thor (andreaslithor@gmail.com)

Version: 2 Date: 06 Sep 2017

Author’s response to reviews:

Dear Editor and reviewer,

We are thankful for your comments and have adapted our manuscript following your recommendations. Each of the eight suggestions are easily tracked in red and green in the revised manuscript. Initially, the title has been changed and thereafter each change can be found throughout the paper.

We hope these changes can make the paper more suitable for publication and we remain at your disposal further.

Kind regards,

Andreas Thor
Corresponding author